Author's response to reviews

Title: Quantitative assay for the detection of the V617F variant in the Janus kinase 2 (JAK2) gene using the Luminex xMAP technology

Authors:

François W Paradis (francois.paradis@ecogene21.org)
Raynald Simard (raynalds@videotron.ca)
Daniel Gaudet (daniel.gaudet@umontreal.ca)

Version: 4 Date: 15 February 2010

Author's response to reviews:

Corrected capital letters in the title and one typo p.4 fibrosis changed for fibrosis.